Author's response to reviews

Title: Preliminary development of a scale to measure stigma relating to sexually transmitted infections among women in a high risk neighbourhood

Authors:

Melanie LA Rusch (mrusch@ucsd.edu)
Jean A Shoveller (jean.shoveller@ubc.ca)
Susan Burgess (burgnesb@interchange.ubc.ca)
Karen Stancer (karen.stancer@shaw.ca)
David M Patrick (david.patrick@bccdc.ca)
Mark W Tyndall (mtyndall@cfenet.ubc.ca)

Version: 4 Date: 15 September 2008

Author's response to reviews:

Dear Editor,

Please find attached the editorial revisions requested for publication of our manuscript in BMC Women's Health.

The author qualifications and affiliations have been amended. Reference 22, which was previously in press, is now available on PubMed and has been revised in the reference list. The tables have been re-formatted and all vertical lines have been removed. The tracked changes have been removed, and the reference that was extracting incorrectly should be fixed.

Thank you very much and please advise if further changes are needed.

Regards,
Melanie Rusch